Author's response to reviews

Title: Socio-economic factors associated with delivery assisted by traditional birth attendants in Iraq, 2000

Authors:

Seter Siziya (ssiziya@yahoo.com)
Adamson S Muula (muula@email.unc.edu)
Emmanuel Rudatsikira (erudatsikira@llu.edu)

Version: 4 Date: 31 March 2009

Author's response to reviews:

Dear Editor,

We were asked to do the following:

1. Report Authors' contributions in journal format and;
2. Remove Table and figure placement text.

Both these have been done and the manuscript has been checked.

adamson muula